Author's response to reviews

Title: Acceptability and effectiveness of opportunistic referral of smokers to telephone cessation advice from a nurse: a randomised trial in Australian general practice

Authors:

Jane M Young (jyoung@email.cs.nsw.gov.au)
Seham Girgis (sgirgis@med.usyd.edu.au)
Tracey A Bruce (tbruce@thegeorgeinstitute.org)
Melissa Hobbs (melissa.hobbs@latrobe.edu.au)
Jeanette E Ward (jeanetteward@rogers.com)

Version: 5

Date: 22 February 2008